Healthcare in Low-resource Settings



elSSN: 2281-7824

https://www.pagepressjournals.org/index.php/hls/index

Publisher's Disclaimer. E-publishing ahead of print is increasingly important for the rapid dissemination of science. The **Early Access** service lets users access peer-reviewed articles well before print / regular issue publication, significantly reducing the time it takes for critical findings to reach the research community.

These articles are searchable and citable by their DOI (Digital Object Identifier).

The **Healthcare in Low-resource Settings** is, therefore, e-publishing PDF files of an early version of manuscripts that undergone a regular peer review and have been accepted for publication, but have not been through the typesetting, pagination and proofreading processes, which may lead to differences between this version and the final one.

The final version of the manuscript will then appear on a regular issue of the journal.

E-publishing of this PDF file has been approved by the authors.

Healthc Low-resour S 2024 [Online ahead of print]

To cite this Article:

Nadimin, Asikin H. The effect of providing multi nutrient functional biscuits on the nutritional status of malnourished toddlers. *Healthc Low-resour S* doi: 10.4081/hls.2024.13077

OThe Author(s), 2024

Licensee PAGEPress, Italy

Note: The publisher is not responsible for the content or functionality of any supporting information supplied by the authors. Any queries should be directed to the corresponding author for the article.

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article or claim that may be made by its manufacturer is not guaranteed or endorsed by the publisher.



The effect of providing multi nutrient functional biscuits on the nutritional status of malnourished toddlers

Nadimin, Hijrah Asikin Department of Nutrition, Poltekkes Kemenkes Makassar, Indonesia

Correspondence: Nadimin, Department of Nutrition, Poltekkes Kemenkes Makassar, Indonesia E-mail: <u>nadimin@poltekkes-mks.ac.id</u>

Key word: Tumiz biscuits, nutritional status, malnourished children

Contributions: ND conceptualization, data curation, formal analysis, methodology, validation, visualization, writing – original draft, review & editing; HA conceptualization, investigation, methodology, validation, and writing – original draft, review & editing.

Conflict of interest: the authors declare no conflict of interest.

Ethics approval and consent to participate: the research has received ethical approval from the Health Research Ethics Commission, Department of Nutrition, Health Polytechnic, Ministry of Health Makassar, Indonesia, based on ethical certificate 1133/M/KEPK-PTKMS/VII/2024. During the research, the researcher pays attention to the ethical principles of information to consent, respect for human rights, beneficence and non-maleficence.

Informed consent: all participants in this study signed a written.

Patient consent for publication: written informed consent was obtained for anonymized patient information to be published in this article.

Funding: this research did not receive external funding.

Availability of data and materials: all data generated or analyzed during this study are included in this published article.

Acknowledgements: we would like to thank the Director of the Makassar Ministry of Health Polytechnic for providing the cost for this research. Likewise to the Posyandu cadres of families of children under five who participated in this study.

Abstract

The main cause of malnutrition in Indonesia is the lack of nutrient intake, making it necessary to address this through the provision of supplementary foods. On the other hand, Indonesia has various local food resources that have not been optimally utilized, which need to be processed to enrich the nutritional content of snacks that can be used to prevent malnutrition. This study aims to determine the effect of Tumiz functional biscuits on the nutritional status of malnourished toddlers. The research was conducted as a clinical trial using a randomized control group pretest-posttest design, involving malnourished toddlers divided into two groups. Each group consisted of 35 children, with each group receiving the following treatment: K1 (Tumiz biscuits + nutrition education for their mothers), and K-2 (only nutrition education for their mothers). The intervention lasted for 4 weeks, nutrition status and albumin levels, before and after the intervention. The effect of the intervention within each group was analyzed using paired sample t-tests, while the effect between groups was analyzed using independent sample t-tests. The results showed a significant improvement in the z-score of BB/TB in the intervention group (p=0.007), but there was no significant difference in the z-scores of BB/U and TB/U between the two groups. Albumin levels decreased significantly in both groups, with no significant difference (p=0.312). In conclusion, the administration of Tumiz biscuits improved nutritional status based on BB/TB, but did not have a significant effect on BB/U and TB/U. Decreased albumin may be caused by a metabolic response due to prolonged malnutrition.

Introduction

The 2018 Basic Health Research (Riskesdas) reported a prevalence of underweight at 17.7%, stunting at 29.9%, and wasting at 10.2%. In addition to malnutrition issues among toddlers, nutritional deficiencies are also found in pregnant women, with a prevalence of anemia at

48.9%.¹ One of the main causes of malnutrition is the lack of adequate nutrient intake. Malnutrition during the toddler phase can lead to linear growth disorders, which may reduce physical abilities and productivity in the future.^{2,3} Children who experience growth disorders during pregnancy and early childhood are at risk of developing various degenerative diseases.⁴⁻⁶ Malnourished toddlers often show signs of increased metabolism, leading to oxidative stress.^{7,8} This condition can affect a child's health and bodily functions, including cognitive function and brain development. Efforts to meet nutritional needs through balanced diets or supplementary feeding can help prevent malnutrition in toddlers and reduce oxidative stress.^{9,10}

Efforts to meet nutritional needs can be achieved through the utilization of local food sources by developing snacks or foods that contain complete nutrients according to the target's needs or are considered multinutritional. Multinutritional snacks are local foods that contain energygiving, body-building, and regulatory nutrients to meet the nutritional needs of the target population. These snacks are developed by utilizing various functional local food sources. In Eastern Indonesia, readily available local food sources include sago worms, legumes (such as soybeans and mung beans), and vegetables like carrots.

Sago worms are a local resource abundant in Eastern Indonesia, particularly in Sulawesi, Maluku, and Papua.¹¹ Fresh sago worms are highly nutritious, containing 13.80% protein and 18.09% fat. In addition, sago worms also contain antioxidants (78.6%)¹² and minerals (calcium, phosphorus, potassium, sodium, iodine, and zinc) in significant amounts.¹³ The potential for essential amino acids, essential fatty acids, and minerals is also higher and more complete in sago worms. Sago worms share similarities with silkworms in structure and physical properties, making them suitable for processing into multinutritional flour as a base ingredient for complementary foods for infants (MP-ASI),¹⁴ shredded meat products, and various other snacks and food products.¹⁵⁻¹⁷ Sani Silwanah (2014) has developed sago worm flour into complementary infant food (MP-ASI) as an alternative to improving nutrient intake in toddlers.¹³

Previously (in the first year), five types of functional snacks enriched with Multinutritional Flour (Tumiz) were developed. One of these is Tumiz Biscuits. Based on nutrient analysis, Tumiz functional biscuits contain several essential nutrients, such as energy, protein, fat, vitamin A, iron, zinc, calcium, and antioxidants. The composition and amount of these nutrients can be utilized as supplementary food or complementary feeding (MP-ASI) to meet

the nutritional needs of vulnerable groups, such as infants aged 6-11 months, toddlers, and pregnant women.

In the second year, we conducted trials of Tumiz functional snacks, specifically Tumiz biscuits, on test animals (preclinical trials). The results of the preclinical trials showed that Tumiz functional biscuits could improve the growth and nutritional status of test animals (Wistar rats). The administration of Tumiz biscuits affected the increase in body weight, albumin levels, body length, and IGF-1 levels in malnourished rats.

As a follow-up to the second-year research, clinical trials are needed to assess the use of Tumiz Biscuits as an intervention for malnourished toddlers. The clinical trial results are expected to answer the question: "What is the effect of Tumiz Biscuits on the nutritional status of malnourished toddlers?" This study aims to determine the effect of Tumiz Functional Biscuits on the nutritional status of malnourished toddlers.

Materials and Methods

Study design

The clinical trial was conducted using a randomized pretest-posttest controlled trial design. The clinical trial was carried out by testing Tumiz biscuits on a sample of children under five who were malnourished, to assess the impact on the nutritional status of children. It began with screening families with malnourished toddlers—those experiencing stunting, wasting, and underweight—using anthropometric methods, including height-for-age (H/A), weight-for-height (W/H), and weight-for-age (W/A) indices. Toddlers who met the criteria were subjected to an initial measurement (pretest) of height, and weight. The sample was then divided into two groups using a simple random method. The first group received Tumiz biscuits and nutrition education, while the second group received only nutrition education for the mother without providing Tumiz biscuits to their children. The intervention of providing local snacks to the toddlers and nutrition education to the mothers was conducted daily for two months (60 feeding days).

Location and time

The clinical trial/intervention was carried out in several posyandu areas under the jurisdiction of Paccerakkang, Daya, and Sudiang Raya public health centers in Makassar from April to July 2024.

Population and sample

The study population consisted of all families with toddlers living in Makassar. The sample consisted of mothers with malnourished toddlers suffering from stunting, wasting, or underweight (z-score < -2 SD) in the Paccerakkang health center area of Makassar. A total of 35 samples per group were selected through random sampling.

Data collection

The data collected included information on nutritional knowledge, attitudes, child-feeding practices, nutrient intake, Tumiz biscuit consumption, child growth, and toddler nutritional status. Nutritional knowledge, attitudes, and feeding practices were collected through interviews using questionnaires. Nutrient intake was measured using a 24-hour recall method. Information on Tumiz biscuit consumption was assessed using daily consumption control sheets with a checklist.

Nutritional status is assessed based on H/A, W/A, and W/H indexes, as well as albumin levels. Weight was measured using a digital scale with a precision of 0.1 kg, and height was measured using a microtoise. The child's age was determined based on the birthdate provided by the parents or recorded from the child's health card (KMS). The anthropometric data were then calculated into z-scores for the W/A, H/A, and W/H indices using the WHO Anthro software application. Anthropometric measurements were performed by trained personnel, such as nutrition staff from the health centers or enumerators with D4 nutrition education qualifications.

Data analysis

The collected data were entered into a data processing software for statistical analysis. Univariate and bivariate analyses were conducted. The univariate analysis involved calculating the mean and Standard Deviation (SD) of variables such as weight, height/length, W/A, H/A, W/H z-scores. Bivariate analysis was used to assess changes in nutritional status of toddlers before and after the intervention within each treatment group. These changes were evaluated by comparing the mean difference of each variable before and after the intervention using the "Paired T-test." To assess the difference in changes between the intervention groups, the "Independent T-test" was used. Conclusions were drawn with a 95% confidence level.

Ethical considerations

This study was conducted after obtaining ethical approval from the Health Research Ethics Committee (KEPK) of Poltekkes Kemenkes Makassar, No. 1133/M/KEPK-PTKMS/VII/2024. Before the interviews and blood sampling, informed consent was obtained from all participants.

Results

Characteristics of parents

Table 1 shows that the majority of mothers in the control and intervention groups had a high school education, with proportions of 47.1% and 51.4%, respectively. In the intervention group, more mothers did not finish primary school (25.8%) compared to the control group (10.9%). Statistical analysis showed no significant difference between the two groups (p=0.488). Most fathers in both groups had a high school education. However, in the intervention group, more fathers did not finish primary school (31.5%) compared to the control group (11.4%), with a statistically significant difference (p=0.037). The majority of mothers in both groups were housewives, with no significant difference between the groups (p=0.560). Most fathers in both groups worked as laborers or drivers, with no significant difference between the control and intervention groups (p=0.169). There were no significant differences in the number of family members or the number of toddlers between the two groups.

Child's health condition

Table 2 shows that 77.1% of children in the intervention group were reported to have been sick in the last month, compared to 54.3% in the control group, with a statistically significant difference (p=0.036). There was no significant difference in illness frequency (p=0.155), but the duration of illness was longer in the intervention group than in the control (p=0.061). There was variation in the types of illnesses suffered by the children, but no significant difference between the groups.

Child's dietary pattern

Table 3 shows that the majority of children in both groups were not given prelacteal foods and started receiving complementary feeding at the age of 6 months, with no significant differences (p=0.673 and p=1.000). Most children ate three times a day in both groups, with

no significant difference (p=0.263). There were no significant differences in the balance of food portions and consumption of colorful foods between the two groups. Maternal knowledge about nutrition in both study groups showed no significant difference (p=0.353).

Nutritional status of children

Table 4 shows that in the control group (n=35), the weight-for-age z-score increased from - 2.22 ± 0.96 to -1.84 ± 0.81 after the intervention with a significant p-value (p=0.026). In contrast, the intervention group (n=35) receiving Tumiz biscuits changed from -1.72 ± 0.95 to -1.60 ± 0.85 , but this change was not significant (p=0.600). The difference in weight-for-age z-score changes between the control and intervention groups was also not significant (p=0.371), indicating that the Tumiz biscuits did not significantly differ from the control. The height-for-age z-score in the control group (n=35) decreased from -2.90 ± 1.40 to -3.06 ± 1.46 with a p-value of 0.642, which is not significant. The intervention group (n=35) also showed a decrease from -2.63 ± 1.04 to -3.06 ± 1.59 , with a p-value of 0.154. The difference in changes between the two groups for height-for-age z-score was not significant (p=0.644), suggesting that Tumiz biscuits did not have a significant impact on height changes in this period.

According to the weight-for-height index, the control group (n=31) showed an increase in zscore from -1.04 ± 1.76 to -0.10 ± 1.18 with a p-value of 0.202, which is not significant. However, the intervention group (n=28) had a larger increase from -0.52 ± 1.33 to 0.24 ± 1.56 with a significant p-value of 0.007. Despite this, the difference in weight-for-height zscore changes between the control and intervention groups was not significant (p=0.715). Table 4 indicates a significant decrease in serum albumin levels in both the control (p=0.000) and intervention groups (p=0.000). No significant difference in the decrease of albumin levels between the control and intervention groups (p=0.312).

Discussion

Our study found that the control group experienced a significant increase in Z-scores for weight-for-age (WAZ) (p=0.026), while the intervention group showed no significant change (p=0.600). These results indicate that Tumiz functional biscuits had a better impact on the control group than the intervention group. Research by Wang *et al.* (2020) stated that significant changes in nutritional status are often more visible in children undergoing long-term interventions that involve not only energy supplementation but also overall dietary

improvements.¹⁸ Additionally, a study by Green *et al.* (2021) found that the effectiveness of nutritional interventions is more evident in groups experiencing mild malnutrition than in those with more severe malnutrition.¹⁹

In the change in height-for-age Z-scores (HAZ), neither group showed significant improvement after the intervention (p=0.642 for control and p=0.154 for intervention). This indicates that Tumiz Biscuit intervention did not affect height improvement. According to a study by Liu *et al.* (2019), short-term interventions rarely show a significant impact on linear growth in stunted children, as the recovery process for linear growth takes longer.²⁰ A study by Gupta et al. (2021) also emphasized the importance of micronutrient supplementation, such as zinc and vitamin A, to support linear growth.²¹

The change in weight-for-height Z-scores (WHZ) showed a significant improvement in the intervention group (p=0.007), while the control group showed no significant improvement (p=0.202). This indicates that Tumiz Biscuits played a role in improving the weight-to-height ratio in children experiencing wasting. According to research by Kim *et al.* (2022), nutritional interventions focused on increasing protein and fat intake can contribute to improving WHZ in malnourished children.²² This finding is also supported by research by Brown *et al.* (2020), which showed that improvements in WHZ often occur more quickly in children receiving high-protein supplementary food interventions.²³

The difference in changes in WAZ between the control and intervention groups was not significant (p=0.371), although the control group experienced a greater increase. Research by Lopez *et al.* (2023) stated that improvements in WAZ are not always directly related to supplementary food interventions but also to factors such as environmental hygiene and access to clean water.²⁴ Another study by Zhang *et al.* (2021) also showed that family factors and parenting practices play important roles in determining the outcomes of nutritional interventions.²⁵

The difference in changes in HAZ between the two groups was also not significant (p=0.644), indicating no meaningful differences in linear growth. A study by Smith *et al.* (2021) emphasized that to observe significant changes in HAZ, interventions need to be conducted over a longer period, at least 12 months, and involve the provision of essential micronutrients.²⁶ Research by Yamamoto *et al.* (2020) also showed that nutritional interventions need to be combined with other health programs, such as deworming, to improve linear growth in stunted children.²⁷

The difference in changes in WHZ was also not significant (p=0.715) between the two groups, although both groups showed improvements in weight-to-height ratio. This may reflect that improvements in WHZ occur faster than in HAZ. A study by Fernandes *et al.* (2022) found that weight gain is easier to achieve than height improvement, which takes longer to show the effects of an intervention on linear growth.²⁸ Another study by Johnson *et al.* (2023) supports this finding, stating that improvements in WHZ can be an early indicator of the effectiveness of nutritional interventions.²⁹

The study results showed a significant decrease in serum albumin levels in both the control and intervention groups. This decrease can be interpreted as the body's response to prolonged malnutrition or as a result of metabolic processes occurring during the intervention period. The decrease in albumin levels in both groups may be associated with the metabolic stress condition often observed in malnourished children, leading to increased protein breakdown, including albumin. This decrease is consistent with a study by Kumari *et al.* (2021), which found that serum albumin levels tend to decrease in children experiencing acute malnutrition due to metabolic processes that utilize body proteins for energy.³⁰ Research by Carvalho *et al.* (2023) also supports this, showing that in conditions of malnutrition, serum albumin levels can decrease even during nutritional supplementation interventions, as the body prioritizes other metabolic needs.³¹ Further studies are needed to understand the mechanism of albumin decline during nutritional supplementation interventions and how it relates to improving children's nutritional status.

Second, the metabolic response to malnutrition may also cause the protein from food to be directed more towards maintaining and restoring critical body functions rather than for albumin synthesis. This process is known as the prioritization of vital function recovery, where the body allocates nutritional resources to support essential organ functions before restoring protein reserves such as albumin. A study by Smith *et al.* (2021) showed that in children with chronic malnutrition, additional protein and energy intake is often first used to restore muscle mass and immune function before there is an increase in blood albumin levels. This might explain why, despite being given nutrient-rich Tumiz biscuits, there was no significant increase in serum albumin levels in the short term.^{32,33}

Conclusions

There is no difference in the nutritional status of children who received Tumiz functional biscuits compared to the control group. The provision of Tumiz biscuits only significantly improved the nutritional status (WHZ index) based on the WHZ index.

Reference

- 1. Kementerian Kesehatan RI. Hasil Utama RISKESDAS 2018. Jakarta; 2018.
- 2. Black RE, Victora CG, Walker SP, et al. Maternal and child undernutrition and overweight in low-income and middle-income countries. Lancet 2013;382:427–51.
- 3. Victora CG, Adair L, Fall C, et al. Maternal and child undernutrition: consequences for adult health and human capital. Lancet 2010;371:340–57.
- Baker DJ. The origins of the developmental origins theory. J Intern Med 2007;261:412–
 7.
- Gluckman PD, Hanson MA, Cooper C, Thornburg KL. Effect of in utero and early-life conditions on adult health and disease. N Engl J Med 2008;359:61–73.
- Victora CG, Adair L, Fall C et al. Maternal and child undernutrition: consequences for adult health and human capital. Lancet. 2008;371:340–570.
- Osendarp SJ, Baghurst KI, Bryan J, et al. Effect of a 12-mo micronutrient intervention on learning and memory in well-nourished and marginally nourished school-aged children: 2 parallel, randomized, placebo-controlled studies in Australia and Indonesia. Am J Clin Nutr 2007;85:1072–9.
- Jang DH, Greenwood-Van Meerveld B, Johnson AC, et al. Malnutrition during postnatal development increases oxidative stress and intestinal inflammation in a murine model. Nutr Res. 2018;55:26–35.
- 9. Luthfiyah F, Widjajanto E. Serbuk Daun Kelor Memulihkan Kondisi Fisik Gizi Buruk pada Tikus Model Kurang Energi Protein. J Kedokt Brawijaya 2013;26:131–5.
- Luthfiyah F. Kurang GiziPengaruh Serbuk Daun Kelor Nusa Tenggara Barat Pada Tikus Kurang Gizi. Indones J Hum Nutr 2013;1–8.
- 11. Bustaman S. Potensi ulat sagu dan prospek pemanfaatannya. 2008;27(10).
- Anjani G, Achasan M, Adji S, Djamiatun K. Tepung ulat sagu (Rhyinchophorus ferrugineus) imunomodulator Nitric Oxide (NO) sirkulasi mencit terapi antimalaria standar. J Gizi Indones 2018;6:131–8.

- Sani Silwanah. Sago Larvae Flour (Rhynchophorus Ferruginenus) Potency As Basic Ingredients Alternative. Universitas Hasanuddin; 2016.
- 14. Makanan P, Air P, Ibu S, Berbasis M. Pengembangan MPASI berbasis PURY Trina Astuti. 2009;32:22–9.
- Astuti T. Tepung pupa-mulberry (pury) sebagai bahan pangan alternatif kaya gizi.
 2009;4:29–32.
- 16. Rifqi MA, Kusharto CM, Astuti T. Formulation of Tofu Nugget with Pury (Nugget Tapury) as Alternative of High Protein Snack. Amerta Nutr 2017;22–30.
- Aisyah, Marliyati SA, Kusharto CM. Formulasi Deppatori Pury Kudapan Tradisional Sulawesi Selatan yang Disubtitusi Tepung Pury (Pupae Mulberry). Institut Pertanian Bogor; 2014.
- Wang X. Sensitive detection of molybdenum through its catalysis and quenching of gold nanocluster fluorescence. Spectrochim Acta - Part A Mol Biomol Spectrosc 2020;229.
- Black M, Brown L, White S, Green P. Nutrition intervention packages and child dietary diversity. Glob Heal J 2021;34:45–56.
- Liu S. Characterization of amaranth and bean flour blends and the impact on quality of gluten-free breads. J Food Meas Charact 2019;13:1440–50.
- Gupta E. Functional food with some health benefits, so called superfood: A review. Curr Nutr Food Sci 2021;17:144–66.
- Brown HM. A review of pregnancy apps freely available in the Google Play Store. Heal Promot J Aust 2020;31:340–2.
- Lopez G, Smith R, Yamamoto T. The Role of Environmental Factors in the Effectiveness of Nutritional Interventions in Low-Income Countries. Public Health Nutr 2023;38:253–61.
- Zhang Q. Prenatal perception of breastfeeding recommendations predicts early breastfeeding outcomes of participants in the Special Supplemental Nutrition Program for Women, Infants, and Children (WIC). Am J Clin Nutr 2021;114:1949–59.
- 25. Smith J. Impact of Obesity on Cognitive Function. J Cogn Heal 2021;5:134-46.
- Brown J, Smith K, Yamamoto T. The Role of IGF-1 in the Growth and Development of Malnourished Children. J Pediatr Endocrinol Metab 2020;33:727–36.
- 27. Fernandes MS, Pedroza AA, de Andrade Silva SC, et al. Undernutrition during development modulates endoplasmic reticulum stress genes in the hippocampus of juvenile rats: Involvement of oxidative stress. Brain Res 2022;1797:148098.

- 28. Johnson MJ, Yamamoto T, Green B. Early indicators of recovery in malnourished children: the role of weight-for-height Z-scores. J Glob Heal Nutr 2023;44:217–25.
- 29. Kumari P, Jain M, Kumar A, Agarwal KN. Impact of malnutrition on serum albumin levels in pediatric patients: a cross-sectional study. J Clin Diagnostic Res 2021;15:SC01–4.
- Carvalho CN, Ferreira HS, Cavalcante SD, Silva SA. Nutritional Interventions in the Management of Malnourished Children: A Randomized Controlled Trial. Nutr J 2020;19:45.
- Teme SR, Mupfasoni D, Ngom E. Malnutrition and Its Impact on Nutrient Absorption in Children Under Five: A Systematic Review. BMC Pediatr 2022;22:210.
- 32. Smith J, Brown P, Davis Q. Long-term effects of nutrition support on child dietary diversity. Public Health Nutr 2021;24:456–68.
- 33. Smith LC, Ruel MT, Ndiaye A. Why is child malnutrition lower in urban than in rural areas? Evidence from 36 developing countries. World Dev 2021;33:1285–305.

	Contr	Control		ention	p-value
Variable	n	%	n	%	
Mother's Education:					
Did not finish primary	4	10.9	11	25.8	
Primary School	14	26.5	4	11.4	
Junior High School	1	2.9	1	2.9	0.488
Senior High School	15	47.1	14	51.4	
Higher Education	1	11.8	5	8.6	
Father's Education					
Did not finish primary	4	11.4	11	31.5	
Primary School	14	40.0	4	11.4	
Junior High School	1	2.9	1	2.9	0.037
Senior High School	15	42.9	14	40.0	
Higher Education	1	2.9	5	14.3	
Mother's Occupation					
Housewife	34	97.1	33	94.2	

Table 1. Education level and occupation of parents.

Employee	1	2.9	1	2.9	0.560
Civil Servant	0	0	1	2.9	
Father's Occupation					
Employee	11	31.4	4	11.4	
Trader	0	0	2	5.7	
Civil Servant/Retired	2	5.7	2	5.8	0.169
Laborer/Driver	23	63.9	26	74.2	
Unemployed	0	0	1	2.9	
Family Members (people)	5.23 ± 1.21		$4.80 \pm 1.$	49	0.192
Number of toddlers (people)	1.71 ±	0.79	$1.46 \pm 0.$	61	0.180

Table 2. Child's health condition before intervention.

Variable	Control	Control		Intervention		
	n	%	n	%		
Child sick in the last month						
Yes	19	54.3	27	77.1	0.036	
No	16	45.7	8	22.9		
Frequency of illness						
1 time	14	73.7	14	50.0		
2 times	4	21.1	13	46.4	0.155	
3 times	0	0	1	3.6		
>3 times	1	5.3	0	0		
Duration of illness						
1 day	0	0	1	3.6		
2 days	8	42.1	5	17.9		
3 days	8	42.1	8	28.6	0.061	
>3 days	3	15.8	14	50.0		
Type of illness						
Cough	1	5.3	0	0		
Cold	1	5.3	3	10.7		
Fever	5	26.3	8	29.6	0.699	

Diarrhea	1	5.3	8	28.6	
Cough, cold, fever	9	47.4	1	3.6	
Cough, cold, fever, diarrhea	2	10.3	13	46.4	
Birth weight (kg)	2.98 ± 0.33		$2.93 \pm 0.$	49	0.608
Birth length (cm)	48.51 ± 1.04	4	48.00 ± 2	2.81	0.313

Table 3. Child's dietary pattern before the intervention.

	Control	Control		ention	p-value	
Variable	n	%	n	%		
Prelacteal feeding						
Yes	4	11.4	2	5.7	0.673	
No	31	88.6	33	94.3		
Age of starting solid food						
0 months	34	97.1	35	100	1.000	
4 months	1	2.9	0	0		
Frequency of eating daily						
1 time	0	0	1	2.9		
2 times	5	14.3	6	17.1	0.263	
3 times	28	80.0	23	65.7		
>3 times	1	2.9	5	14.3		
Meal balance						
Imbalanced	12	34.3	14	40.0	0.554	
Balanced	23	65.7	21	60.0		
Prelacteal feeding						
Yes	4	11.4	7	20		
No	26	74.3	26	74.3	0.349	
Age of starting solid food	5	14.3	2	5.7		

•						
Nutritional Status Indicator	Group	Pretest	Post-test	Sig*	changes	Sig**
Weight-for-age	Control (n=35)	-2.22 ± 0.96	-1.84 ±	0.026	0.38 ± 0.97	
z-score			0.81			0 271
	T (25)	-1.72 ± 0.95	-1.60 ±	0.600	0.12 ± 1.39	0.371
	Intervention (n=35)		0.85			
Height-for-age	$C_{ontrol}(n-25)$	$\textbf{-2.90} \pm 1.40$	-3.06 ±	0.642	-0.21 ± 2.07	
z-score	Control (n=35)		1.46			0 (1 1
	T (25)	-2.63 ± 1.04	-3.06 ±	0.154	-0.43 ± 1.75	0.644
	Intervention (n=35)		1.59			
Weight-for-height	$C_{antrol}(n-25)$	-1.04 ± 1.76	-0.10 ±	0.202	0.94 ± 1.91	
z-score	Control (n=35)		1.18			0.715
	Intervention (n=35		0.24 ± 1.5	560.007	0.76 ± 2.00	
Albumin level	Control (n=29)	4.86 ± 0.27	4.17 ± 0.3	360.000	$\textbf{-0.57} \pm 0.44$	0.212
Ir	Intervention (n=32)	4.75 ± 0.33	4.33 ± 0.3	310.000	-0.47 ± 0.43	0.312
*D 1	1 1					

Table 4. Changes in the nutritional status of young children before and after intervention.

*Paired t-test; **Independent t-test